Author's response to reviews

Title: A cross-sectional survey to investigate the quality of care in Tuscan (Italy) nursing homes: the structural, process and outcome indicators of nutritional care.

Authors:

Guglielmo Bonaccorsi (guglielmo.bonaccorsi@unifi.it)
Francesca Collini (francesca.collini@ars.toscana.it)
Mariangela Castagnoli (mariangela.castagnoli@virgilio.it)
Mauro Di Bari (mauro.dibari@unifi.it)
Maria Chiara Cavallini (mariachiara.cavallini@unifi.it)
Nicoletta Zaffarana (nicolettazaффarana@alice.it)
Pasquale Pepe (pasquale_pepe1@virgilio.it)
Alessandro Mugelli (alessandro.mugelli@unifi.it)
Erslia Lucenteforte (ersilia.lucenteforte@unifi.it)
Alfredo Vannacci (alfredo.vannacci@unifi.it)
Chiara Lorini (chiara.lorini@unifi.it)

Version: 6
Date: 27 April 2015

Author's response to reviews: see over
Florence, 27 April 2015

Object: Submission of the revised manuscript entitled “A cross-sectional survey to investigate the quality of care in Tuscan (Italy) nursing homes: the structural, process and outcome indicators of nutritional care.” (Authors: Guglielmo Bonaccorsi, Francesca Collini, Mariangela Castagnoli, Mauro Di Bari, Maria Chiara Cavallini, Nicoletta Zaffarana, Pasquale Pepe, Ersilia Lucenteforte, Alfredo Vannacci, Chiara Lorini)

Dear Editor, Dear Reviewers,

thank you for your comments, that have allowed us to improve our paper. Below the point-by-point response to the concerns. The manuscript has been copyedit to improve the style of written English. Changes in the main test are in red.

We look forward to the referees’ new evaluation, in the hope of a positive reply as to acceptability for publication.

Chiara Lorini and the Co-Authors

Response to reviewer 1 (Felipe Sandoval)
The mistakes have been corrected

- Line 68-69: Capitalizing regional and national seems strange. The text has been changed.
- Line 70: "...care to in NHs". Please use the correct preposition. The text has been changed.
- Line 77: Malnutrition strongly impacts THE quality; or Malnutrition has a strong impact ON the quality... The text has been changed.
- Line 181 When THE data present. The text has been changed.
- Line 182: might NOT be adequate. The text has been changed.
- Line 184 specially>especially. The text has been changed.

Response to reviewer 2 (Jiro Okochi)
The mistakes have been corrected

- Line 315 whit the help of should be “with the help of “. The text has been changed.
- Line 348 Please omit “(confirm or not)” from the sentence. The text has been changed.